Author's response to reviews

Title: Traumatic bone cyst of the mandible in a case of Langer-Giedion syndrome

Authors:

Shakil Ahmed Nagori (drshakilnagori@gmail.com)
Anson Jose (ansonjoseaj@gmail.com)
Bhaskar Agarwal (bhaskar.agg@gmail.com)
Krushna Bhatt (bhattkrushna@gmail.com)
Ongkila Bhutia (dr_ongkila@rediffmail.com)
Ajoy Roychoudhury (ajoyroy@hotmail.com)

Version: 2 Date: 3 May 2014

Author's response to reviews: see over
Saturday, May 03, 2014.

To,
Michael Kidd,
Editor-in-chief,
Journal of medical case reports.

Subject: Submission of manuscript entitled “Traumatic bone cyst of the mandible in a case of Langer-Giedion syndrome”.

Dear Sir,

Please find attached the manuscript “Traumatic bone cyst of the mandible in a case of Langer-Giedion syndrome” for consideration of publication in your esteemed journal. Through our paper we describe a case of Langer-Giedion syndrome which is an extremely rare disorder. Oral and maxillofacial manifestations of such patients are seldom discussed in literature. We present, to the best of our knowledge, the first case of any maxillofacial pathology in this syndrome, while discussing other oral-facial and general signs in these patients.

Also, on behalf of all the contributors I will act and guarantor and will correspond with the journal from this point onward. The authors would like to state that the manuscript is original, and it, or any part of it, has not been and will not be submitted elsewhere for publication; each author has contributed to, read, and approved the manuscript; and none of the authors has any conflict of interest, financial or otherwise.

We hereby transfer, assign, or otherwise convey all copyright ownership, including any and all rights incidental thereto, exclusively to the journal, in the event that such work is published by the journal.

Yours sincerely

(Dr. Ajoy Roychoudhury)